Author's response to reviews

Title: EVALUATION OF A KNOWLEDGE TRANSFER SCHEME TO IMPROVE POLICY MAKING AND PRACTICES IN HEALTH PROMOTION AND DISEASE PREVENTION SETTINGS IN FRENCH REGIONS: A REALIST STUDY PROTOCOL

Authors:

Linda Cambon (linda.cambon@ehesp.fr)
Audrey Petit (Audrey.petit@ehesp.fr)
Valery Ridde (valery.ridde@umontreal.ca)
Christian Dagenais (christian.dagenais@umontreal.ca)
Marion Porcherie (Marion.porcherie@ehesp.fr)
Jeanine Pommier (jeanine.pommier@ehesp.fr)
Christine Ferron (Christine.ferron@fnes.fr)
Laetitia Minary (L.MINARY@chru-nancy.fr)
François Alla (francois.alla@univ-lorraine.fr)

Version: 2 Date: 13 Jun 2017

Author’s response to reviews:

Dear Editor in Chief,

Please find below the point by point response to the comments.

Best regards

Linda Cambon, on behalf of the coauthors.

Commentary 1 :

In Table 1 and 2, please change the label of the column titled 'Collecting ways' to Data collection. We also note that in Table 2 this column is incomplete so please amend accordingly.
Response: The corrections have been made in Tables 1 and 2

Commentary 2:

Please add a sentence to the methods section that you have reported this manuscript in line with the RAMESES II reporting standards for realist evaluation.

Response: The sentence has been added page 6 line 156-157 in methods section

Commentary 3:

Please also take a moment to check our website at http://imps.edmgr.com/ for any additional comments that were saved as attachments.

Response: We saw the « Reference checking » and corrected the 6 references displayed as « not validated ». We didn’t see anything else to do, please let us know if we forgot something.